

May 26, 2014

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 10003-Edited for download.docx).

Title: Prevalence survey and risk factors analysis of gallstone in Uighur and Han

Author: Li Zhu, Akber Eric, Cheng Zhang, Eric Cetin, Kerim Abdiryim

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 10003

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) Please confirm the columns according to the "ESPS-Columns scope note".

Answer: The columns of our study is "Brief Article"

(2) Please provide the city and postcode.

Answer: The city and postcode have been provided.

(3) Please provide the city and postcode.

Answer: The city and postcode have been provided.

(4) Please provide the author contributions. See the format in the attachment file-revision policies.

Answer: The author contributions have been provided.

(5) Please provide the address in detail.

Answer: The address details have been provided.

(6) Please provide the telephone and fax number.

Answer: The telephone and fax number have been provided.

(7) This section should be no more than 20 words.

Answer: The conclusion has been revised.

(8) Please write a summary of less than 100 words to outline the most innovative and important arguments and core contents in your paper to attract readers.

Answer: The core tip has been written.

(9) Please provide all authors abbreviation names and manuscript title here.

Answer: All author abbreviation names and manuscript title have been provided.

(10) Many type errors in the text. Please send the paper to an expert language editing company.

Answer: The errors in the text have been corrected by language-editing experts.

(11) Please update the format of the reference numbers like this one.

Answer: The format of the reference numbers has been updated.

(12) Discussion is too long, Some data are repeated in results and discuss section. Would you like to make a revision?

Answer: We have refined the discussion, and deleted the repeated parts in the results and discussions.

(13) Please write the COMMENTS section at here. See the format in the attachment file-revision policies (High lighted contents).

Answer: We have written the COMMENTS section according to file-revision policies

(14) Please add PubMed citation numbers and DOI citation to the reference list and list all authors. Please revise throughout. The author should provide the first page of the paper without PMID and DOI.

PMID (http://www.ncbi.nlm.nih.gov/entrez/query.fcgi?db=PubMed) DOI
(<http://www.crossref.org/SimpleTextQuery/>)

Answer: We have added the PMID numbers and DOI numbers in English journal articles and added the DOI numbers in Chinese journal articles. But not found the DOI numbers in reference 8, 40 and 48. The papers of 8, 40 and 48 are enclosed.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Li Zhu

Zhu LI, MD

People's Hospital of Xinjiang Uygur Autonomous Region

Urumqi 830000, China

Fax: +86-991-8561562

E-mail: zhuliuygur@163.com